Author's response to reviews

Title: Suppression of low-density lipoprotein oxidation, vascular smooth muscle cell proliferation and migration by a herbal extract of Radix Astragali, Radix Codonopsis and Cortex Lycii

Authors:

Judy Y Chan (judychanyw@hotmail.com)  
Johnny C Koon (johnnykoon@gmail.com)  
Ping-Chung Leung (pingcleung@cuhk.edu.hk)  
Chun-Tao Che (chect@cuhk.edu.hk)  
Kwok-Pui Fung (kpfung@cuhk.edu.hk)

Version: 7 Date: 19 April 2011

Author's response to reviews:

The following points have been revised:
1. One more affiliation has been added to the author list
2. Some missing words in Figure 6A have been added